CARBON PRICING AND INFLATION VOLATILITY



BANCO DE ESPAÑA Eurosistema

Documentos de Trabajo N.º 2231

Daniel Santabárbara and Marta Suárez-Varela

CARBON PRICING AND INFLATION VOLATILITY

Carbon pricing and inflation volatility (*)

Daniel Santabárbara

BANCO DE ESPAÑA

Marta Suárez-Varela

BANCO DE ESPAÑA

Documentos de Trabajo. N.º 2231 October 2022

^(*) We would like to acknowledge Mario Alloza, Ángel Estrada, Filippo Natoli, Florens Odendahl, Javier J. Pérez, Pedro del Río and an anonymous referee as well as other participants at the Banco de España seminar for their valuable comments. We also thank Daniel Montero for his assistance in gathering some of the data that we employed in this paper. All remaining errors are obviously ours. Correspondence to Daniel Santabárbara (daniel. santabarbara@bde.es) and Marta Suárez-Varela (marta.suarez-varela@bde.es).

The Working Paper Series seeks to disseminate original research in economics and finance. All papers have been anonymously refereed. By publishing these papers, the Banco de España aims to contribute to economic analysis and, in particular, to knowledge of the Spanish economy and its international environment.

The opinions and analyses in the Working Paper Series are the responsibility of the authors and, therefore, do not necessarily coincide with those of the Banco de España or the Eurosystem.

The Banco de España disseminates its main reports and most of its publications via the Internet at the following website: http://www.bde.es.

Reproduction for educational and non-commercial purposes is permitted provided that the source is acknowledged.

© BANCO DE ESPAÑA, Madrid, 2022

ISSN: 1579-8666 (on line)

Abstract

Carbon pricing initiatives, designed to increase the relative prices of greenhouse gas-intensive goods and services, could not only push up CPI inflation but also affect its volatility. Existing empirical literature has only found that carbon pricing schemes are generally associated to a transitory effect on the level of inflation. This paper assesses empirically the effects of carbon pricing on inflation volatility for both carbon tax and cap-and-trade schemes (also known as emission trading systems). Our work finds strong evidence that capand-trade schemes are associated with larger volatility in CPI headline inflation, while no significant effect is found in the case of carbon taxes. This effect seems to feed only through the energy component, and does not seem to affect the volatility of core inflation. In addition, we find that under cap-and-trade schemes, both the increase in the underlying price of emissions and the expansion in the activities covered by these initiatives are associated with greater inflation volatility. These findings have important policy implications, given that inflation volatility could complicate the conduct of monetary policy. Since the ambition to mitigate climate change in the years to come is expected to be implemented through broader coverage of carbon pricing, central banks should monitor those developments closely.

Keywords: carbon pricing, emission trading systems, carbon tax, inflation, inflation volatility.

JEL classification: E31, E32, E52, E58, Q48, Q58.

Resumen

Las iniciativas de fijación de precios del carbono, diseñadas para incrementar los precios relativos de los bienes y servicios intensivos en gases de efecto invernadero, no solo podrían incidir en la inflación del IPC, sino también en su volatilidad. La literatura empírica ha encontrado, de manera general, que estas medidas incrementan transitoriamente el nivel de inflación. Este trabajo analiza empíricamente los efectos de las iniciativas de fijación de precios del carbono sobre la volatilidad de la inflación, tanto de los mercados de derechos de emisión como de los impuestos sobre el carbono. Encontramos abundante evidencia de que los mercados de derechos de emisión conducen a una mayor volatilidad de la inflación. El aumento de la volatilidad provendría del componente energético, mientras que no se aprecia un efecto significativo en el caso de la inflación subyacente. Además, los incrementos del precio efectivo de las emisiones y de las actividades cubiertas por estas iniciativas inducen una mayor volatilidad de la inflación. Por todo ello, y dada la creciente ambición para mitigar el cambio climático, estos resultados cobran cada vez más relevancia en términos de política económica, ya que una mayor volatilidad de la inflación puede complicar la gestión y comunicación de la política monetaria.

Palabras clave: precios del carbono, mercados de derechos de emisión, impuesto sobre el carbono, inflación, volatilidad.

Códigos JEL: E31, E32, E52, E58, Q48, Q58.

1 Introduction

Not only climate change but also its mitigation policies may affect business cycle and inflation dynamics. The scientific consensus, as reflected in the findings of the International Panel on Climate Change, points to an increasing concern that, absent a significant reduction in greenhouse gases (GHG) emissions, the process of global warming will lead to devastating and potentially irreversible consequences for the planet (IPCC, 2022). In light of this evidence, many governments have implemented significant mitigation policies and committed to expand their ambition, fundamentally with regard to the reduction of GHG emissions. Besides the physical risks derived from climate change, those policies aimed at the transformation to a low-carbon economy could also be associated with significant transition risks¹, both affecting the macroeconomic outcomes. The realization that climate change and mitigation policies may reshape policy formulations has prompted the need for a better understanding of their consequences for economic policy.

Monetary policy has been no exception. Both climate change and the transition to a lowcarbon economy could affect the delivery of central bank's price stability mandates and the conduct of monetary policy through various channels. That is, to the extent that these structural processes may induce changes in business cycle and inflation dynamics, affect the equilibrium level of the real interest rate, or generate distortions in the financial system (in particular, in credit allocation), they may hinder the transmission of monetary policy. Indeed, in the context of its new monetary policy strategy, the ECB has recently committed to incorporate climate change considerations into the design of monetary policy for these reasons (ECB, 2021). Also, in several recent speeches ECB Executive Board Member Schnabel (2021, 2022) further emphasized the importance of both physical and transition risks as a threat to price and macroeconomic stability and, hence, to the conduct of monetary policy.

Among the policy measures to mitigate climate change, carbon pricing is commonly seen as the most efficient economic instrument for reducing emissions (Hepburn *et al.*, 2020), since it allows internalizing the external costs derived from the production and consumption of GHGintensive goods, by bringing the implicit cost of emissions closer to their social cost (Pigou,

¹These risks derive from the initiatives to mitigate climate change. For instance, to the extent that many of them lead to an increase in fossil fuel prices in the short term, this could affect the income and credit quality of those households and firms that depend more intensively on this source of energy. Moreover, uncertainty about the public policies and the structural transformation to come could influence economic agents' consumption and investment decisions, without ruling out financial markets turbulence episodes. The probability of the realization of these risks and their intensity will depend on how the transition to a low-carbon economy is designed and implemented.

1932). In fact, the presence of carbon pricing initiatives has increased sharply in the last years, not only regarding the number of initiatives implemented but also in geographical terms and in relation to the emissions covered by those systems (World Bank, 2021).

Carbon pricing initiatives are designed to increase the prices for carbon emissions, providing economic agents with an incentive to conserve energy and switch to greener sources. Then, they should lead to higher prices of those goods and services intensive in GHG emissions, pushing up (at least temporarily) inflation. However, the incentives from carbon pricing to increase energy efficiency and foster innovation in low-carbon sources of energy, especially in the electricity sector, combined to the diminishing weight of fossil fuels in the consumption basket act in the opposite direction.² In this regard, most of the empirical literature so far has focused on the effects of carbon prices on the level of inflation, the interaction between inflationary and disinflationary underlying forces and, in some cases, on the persistence of inflation (Konradt and Weder di Mauro, 2021; Mckibbin *et al.*, 2021; Känzig, 2021; Moessner, 2022). These studies show that carbon pricing initiatives generally produce a transitory effect on the level of headline inflation. However, to our knowledge no previous work has dealt with the effect of carbon pricing schemes on the volatility of inflation.

Nevertheless, carbon pricing initiatives could be expected to affect the inflation process differently depending on the design of those initiatives. Indeed, several works and reports have already suggested a relation between some types of carbon pricing systems and price volatility (see, for example, Metcalf (2021) or ECB (2021)). That is because the presuming impact on inflation volatility of the two main carbon pricing initiatives, cap-and-trade schemes (also known as emissions trading schemes (ETS)) and carbon taxes³, are different. Under ETS, a cap is first set on pollution, issuing afterwards a volume of emission allowances compatible with that cap and allowing companies to trade with them and determine a market clearing price, which could be potentially rather volatile. Conversely, under carbon taxes schemes, policy makers set a price on GHG emissions, letting the market determine the amount of pollution consistent with that price. Therefore, carbon prices tend to be relatively stable as long as their tax rates and tax base do not change so often. All in all, while a carbon tax is expected to lead to relatively stable carbon prices, cap-and-trade systems have been found to lead to substantial price volatility (Goulder and Schein, 2013; Metcalf, 2021). This volatility could be translated into producer

 $^{^{2}}$ As argued by Schnabel (2022), in order to limit the consequences of climate change, carbon prices should remain elevated in the medium term, while investments and innovation in decarbonized technologies to substitute them may need some longer time. Therefore, inflationary pressures may be more persistent.

³A carbon tax is a tax on the supply of fuel in proportion to their carbon content.

and consumer prices, especially of those goods and services related to the sectors under the coverage of ETS. This way, these initiatives could have only a temporary effect on the level of headline inflation but a more permanent effect on its volatility.

In this paper, we fill this gap in the literature by assessing empirically the effects on inflation volatility of the two main types of carbon pricing initiatives, that is to say, carbon taxes and ETS systems. Moreover, we consider the impact not only on the volatility of headline CPI, but also of the energy and the core subcomponents. We also analyse the role played by the level of carbon prices and the share of GHG emissions covered by those carbon pricing initiatives. We find that carbon pricing initiatives in the form of ETS are associated with larger CPI headline inflation volatility, an effect that is not found in the case of schemes of the carbon tax type. Our results also suggest that ETS affect the volatility of inflation through the energy component, since energy related goods are the ones that are targeted directly or indirectly by carbon pricing initiatives, while it does not seem to affect core inflation volatility. Finally, we find that, under ETS, expanding the coverage of those initiatives and increasing the underlying price of emissions leads to greater inflation volatility.

As summarised above, climate change mitigation policies could have a wide-ranging impact on the nominal side of the economy, reflected mainly in the inflation level, their volatility and on asset markets prices, which are likely to affect the implementation and transmission of monetary policy (NGFS, 2021; ECB, 2021). If carbon pricing only affects transiently the level of inflation, sharing the features of a negative supply-side shock, credible monetary policy authorities will generally 'look through' this effect. But if the carbon pricing schemes are also found to affect inflation volatility, those relative price shocks complicate the communication and the conduct of the monetary policy. Since the ambition to mitigate climate change in the years to come is expected to be partly instrumented through broader coverage of carbon pricing initiatives, a better understanding of its implications for the inflation process becomes crucial.

The rest of the paper is structured as follows. Section 2 reviews the literature. Section 3 explains the database built including its descriptive statistics. Section 4 deals with the empirical model and the econometric approach considered. Results are discussed in Section 5 and, finally, Section 6 concludes.

2 Literature review

Despite the increasing presence of carbon pricing schemes in the last decades, the literature on their economic impact has remained relatively scarce, and mostly centered on ex-ante (modelbased) evaluations. Regarding the ex-post (estimated) economic impact of carbon pricing, a growing body of research has recently addressed the study of their effects on economic growth (Metcalf and Stock, 2020), industrial and business performance, competitiveness and innovation (Shapiro and Metcalf, 2021; Venmans *et al.*, 2020; Martin *et al.*, 2014, 2016), employment (Martin *et al.*, 2014; Yamazaki, 2017), international trade (Mundaca *et al.*, 2021) and the inequality of income distribution (Elkins and Baker, 2001; Känzig, 2021).

With respect to inflation, although the field is gaining momentum, evidence is still rather limited. In 2014, McKibbin et al. (2014) derived model-based evaluations of the introduction of a carbon tax in the US, pointing to sizable positive and temporary effects on inflation.⁴ Regarding ex-post analysis of the impacts on inflation, a recent study by Konradt and Weder di Mauro (2021) conclude that carbon taxes in Europe and Canada have not been inflationary, and they may even have been deflationary. They stress that the observed inflationary pressures seem restricted to the headline component and do not affect core inflation, which can be attributed to the fact that carbon taxes alter relative prices - by increasing the price of energy - but do not lead to increases in the price of the broader basket of goods and services. Focusing on the euro area, Mundaca et al. (2021) further explore this relationship, finding a positive effect of carbon taxes on inflation, particularly in the first years following the introduction of the carbon tax. The effect is nevertheless found to fade in the medium to long-term. Moreover, the impact on core inflation tends to be negative, also pointing to carbon taxes impacting inflation through the change in relative prices instead of affecting overall price levels. By exploiting high-frequency data and some institutional features of the European carbon market, Känzig (2021) also derive a positive impact of carbon pricing on energy and consumer prices. Indeed, carbon policy shocks are found to account for around one third of the variations in energy prices in the context of the EU ETS. Finally, Moessner (2022) analyses the effects of carbon pricing on the level of headline, food, energy and core inflation for OECD countries, focusing not only on carbon taxes but also on the prices of ETS. She finds that an increase in the price of ETS is indeed inflationary but its effects are circumscribed to energy CPI inflation and fade away in two years. Surprisingly, an increase in carbon taxes is found to push up food CPI inflation only, and it shows no significant

⁴Model-based evaluations of a carbon tax in Spain (Estrada and Santabárbara, 2021) show similar results: the introduction of a carbon tax translates into a temporary increase in inflation. However, they find that what governments would do with carbon tax revenues will shape the response of the economy to the energy transition.

effects on the rest of components.

To date, this emerging body of research has focused on assessing the impact of carbon pricing on the level of inflation. However, to our knowledge the effects of carbon pricing schemes on the volatility of inflation remain largely unexplored. The impact of carbon pricing on inflation volatility should be expected to depend on several factors. First, carbon pricing systems can take the form of both carbon taxes and cap-and-trade systems, with presumed differing impacts on the volatility of inflation. For cap-and-trade schemes, a cap is first set on pollution, issuing afterwards a volume of emission allowances compatible with that cap and allowing companies to determine a market clearing price by selling and buying allowances. On the contrary, under a carbon tax, policy makers directly set a price on CO_2 emissions, letting the market determine the amount of pollution consistent with that price. Therefore, while a carbon tax is expected to lead to more stable carbon prices, cap and trade systems show substantial price volatility (Goulder and Schein, 2013; Metcalf, 2021). This volatility could be translated into producer and consumer prices, especially of those goods and services related to those sectors under the coverage of carbon pricing. However, the introduction of carbon price initiatives could also reduce the weight of fossil fuels in the CPI consumption basket (Andersson, 2019), which has historically been a source of inflation volatility.

The economic impact of carbon pricing has also proved to be dependent on the level of coverage of the initiatives, on the basis that the pass-through to consumer prices is expected to be higher if the initiative covers a greater proportion of emissions within its jurisdiction (Metcalf and Stock, 2020). The effect of carbon pricing on inflation should also be dependent on the effective price of the initiative (Metcalf and Stock, 2020; Konradt and Weder di Mauro, 2021). There is also evidence that carbon pricing leads to differing impacts according to the component of inflation considered, i.e. headline, energy or core inflation (Konradt and Weder di Mauro, 2021; Mckibbin *et al.*, 2021; Känzig, 2021; Moessner, 2022), which could translate to differential impacts on the volatility of that inflation. Finally, the effects of carbon pricing systems on inflation may depend on some other factors, such as the production structure, and particularly whether the country specializes in high-polluting industries and sectors. That is because these sectors tend to be under the coverage of carbon pricing systems.

Regarding the control variables, the literature has identified a number of domestic and global factors as affecting inflation volatility. First, with respect to the domestic factors, the size of the country and its level of development are usually considered as drivers of inflation volatility (Lane, 1997; Neely and Rapach, 2011; Parker, 2018). Central bank independence and monetary policy 1997; Neely and Rapach, 2011; Parker, 2018). Central bank independence and monetary policy frameworks have also been associated to the capacity of the monetary authority to achieve price stabilization (Cukierman *et al.*, 1992; Bernanke and Gertler, 2001; Klomp and de Haan, 2010). In addition, the domestic level of CPI inflation has also been identified as a factor affecting its volatility (Bleaney and Fielding, 2002). Second, some domestic factors related to the sensitivity of the economy to external conditions also are commonly associated to inflation volatility. For example, trade, its role as commodity exporter, or the *de iure* and *de facto* of capital account liberalization of each economy are all of them related to the reaction of domestic prices to swings in external economic conditions (Romer, 1993; Lane, 1997; Bowdler and Malik, 2017; Aghion *et al.*, 2004). In the same vein, the choice of the exchange rate regime is commonly seen as a mechanism related to how external shocks affect inflation dynamics (Bleaney and Fielding, 2002; Alfaro, 2005; Kamber and Wong, 2020). Third, a strain of the literature has found that inflation dynamics are increasingly explained by global factors, and particularly international commodity prices, rather than by domestic conditions (Fernández *et al.*, 2017; Parker, 2018; Kamber and Wong, 2020).

3 Data

Our sample starts in 2000, after the introduction of the euro⁵ and just before the adhesion of China to the World Trade Organization⁶, two major events affecting the inflation process in advanced economies. We focus our analysis on OECD countries, since inflation in advanced economies has behaved differently than in emerging market economies (Blanchard *et al.*, 2015; Ha *et al.*, 2019). This results in an unbalanced panel of 36 countries for the period 2000-2020. To match the measure of inflation volatility, the resulting panel has biennial frequency.

Our dependent variable on inflation volatility is built from CPI data for headline, core and energy inflation that is retrieved from the OECD Data Portal with monthly frequency. Inflation volatility for these three indices is computed as the standard deviation of year-on-year inflation for each country in windows of 24 months.⁷ Also, for robustness checks we construct some alternative measures of inflation volatility. First, in order to avoid unduly overweighting extreme data points following a common approach in the literature inflation volatility is computed as

⁵That is because a substantial proportion of OECD countries are part of the euro area. The creation of the European Monetary Union in 1999 led to structural changes in monetary policy and in inflation patterns in those countries (Balatti, 2020; Bowdler and Malik, 2005).

⁶China's adhesion to World Trade Organization in 2001 is often seen a milestone on globalization. The globalization of inflation hypothesis argues that the factors influencing inflation dynamics are becoming increasingly global (Forbes, 2019).

⁷The choice of the frequency was motivated by the need to compute inflation volatility relying on a sufficient number of observations. However, it should be noted that it led to the exclusion of Australia and New Zealand, for which only quarterly data is available. Figure A1 in the Appendix depicts the evolution of this measure of inflation volatility.

the log of one plus the standard deviation of inflation (Bowdler and Malik, 2017).⁸ Second, both inflation volatility measures are built using different time windows, 12 and 36 months, respectively.

Information on carbon pricing comes from the Carbon Pricing Dashboard, a comprehensive database with global geographical coverage elaborated by the World Bank, that includes data on either national, subnational and regional (supranational) carbon pricing initiatives.⁹ By processing the information contained in the database, we build some variables of interest for this work. First, in order to capture which jurisdictions have introduced carbon pricing initiatives, we construct a dummy variable that takes value 1 if the country has any type –national, subnational or regional- of carbon pricing implemented for a given year ('Carbon pricing, any'). Second, we distinguish between initiatives of the cap-and-trade (or ETS) and carbon tax types, incorporating two dummy variables: 'ETS, any' and 'Carbon tax, any', that reflect whether the country has implemented any type of ETS or carbon tax, respectively, in a given year. Third, since the effect of carbon pricing on inflation is expected to depend on the underlying carbon price, we compute the weighted average of the real carbon price in each jurisdiction for each year ('Price'), weighing the carbon price of each initiative within the same country by the share of emissions covered by each initiative. Emissions not covered by any initiative are supposed to have a price of zero; therefore, the variable 'Price' could be understood as a sort of real effective price of CO_2 in each country. Carbon prices are expressed in dollar terms and, thus, are deflated by the US CPI. Fourth, the effect of carbon pricing should also be related to the amount of GHG covered by the initiatives in that country. Unluckily the Carbon Pricing Dashboard only offers coverage within the jurisdiction for the last available year, which would significantly reduce the variability and accuracy of the information employed in the analysis. However, the data on global emissions coverage, that is, the percentage of global emissions covered by a given initiative, is offered for each year. Therefore, we are able to compute a proxy of the percentage of emissions covered in each jurisdiction by multiplying global emissions coverage by global emissions themselves -to get total emissions covered by each initiative for a given year- and dividing it by each country's emissions levels that same year.¹⁰ One additional difficulty stems from the fact that,

⁸This allows to down-weight very large readings of inflation that may occur during hyperinflation episodes as well as inflation records close to zero.

⁹See https://carbonpricingdashboard.worldbank.org

¹⁰For the years 2019 and 2020, World Bank does not offer information on GHG emissions, so we assume that the percentage of emissions covered by each carbon pricing initiative in each jurisdiction remains constant after 2018. Although most of the GHG emissions data required are now available for one extra year, 2019, using different sources (such as Climate Analysis Indicator Tool, Potsdam Institute for Climate Impact Research or United Nations Framework Convention on Climate Change), we have opted for preserving the consistence within the World Bank dataset.

for some countries, there may be coexisting national, regional and subnational initiatives that can overlap for a given year in terms of the emissions they cover. Since we cannot rely on accurate information on overlapping emissions for each initiative on an annual basis, we decided to compute two different measures of coverage. The first one considers that the initiatives at different administrative levels are fully overlapping, and it is therefore linked to the initiative with maximum coverage by jurisdiction ('*Coverage-fully overlapping*'). On the contrary, the second measure assumes that the initiatives are not overlapping at all and is obtained by adding up the coverage of each initiative within the country/jurisdiction ('*Coverage-not overlapping*')¹¹.

We also consider some other factors that may affect the relationship between carbon pricing and inflation volatility. In order to account for the production structure, and particularly whether the country specializes in high-polluting industries and sectors, we include the share in value added in each economy of the 20% highest polluting sectors worldwide ('VA share polluting sectors (20%)'). Moreover, countries with carbon pricing initiatives may also rely on more stringent environmental norms, that may impose some implicit abatement costs beside the explicit price for carbon. To control for this fact, the role of those other climate change mitigation policies is captured from the OECD's Environmental Stringency Index¹², which aims at measuring environmental policy stringency internationally over a relatively long time horizon. Finally, to consider the possibility of a structural change after the surge in environmental policies following the Paris Agreement in 2015, we include an additional dummy variable ('Post-2015').

Regarding the usual control variables identified in the literature, we gathered information from different sources. Population, GDP per capita and Trade openness are directly drawn from World Bank's World Development Indicators. For *de iure* central bank independence, we employed the Garriga (2016) dataset¹³. Our proxy for capital flows and financial openness is built as the sum of the absolute value of inflows and outflows in percentage of each country GDP based on IMF International Financial Statistics data. Using data on merchandise exports from World Bank, a country is considered as a primary exporter if exports of food, fuels, and ores and metals imply more than 20% of total merchandise exports. Exchange rate regime classification

¹¹It should be noted that for few countries, such as Finland, Sweden or Norway, and some years may count on coexisting carbon tax and ETS initiatives with relevant coverage. The last measure of coverage may result in levels of higher than 100% for some years. As shown in Section 5.4, our results are robust to the use of the two measures.

¹²The correlation between having carbon pricing initiatives and environmental stringency is indeed not as high as it could be expected, yielding a value of 0.56.

¹³Garriga (2016) uses Cukierman *et al.* (1992) to classify central banks according to several criteria related to the characteristics of the chief executive officer of the bank, policy formulation attributions, objectives, and limitations on lending to the public sector. The resulting index ranges from 0 (lowest) to 1 (highest independence).

follows Reinhart and Rogoff (2004) and Ilzetzki *et al.* (2019).¹⁴ Following Fernández *et al.* (2017) and Kamber and Wong (2020), real commodity prices are accounted through the growth rate in Agricultural, Metal and Minerals, and Energy Commodity Price Indices, deflated by the US CPI, obtained from World Bank's Pinksheet on Commodity Prices.

Table 1 shows main descriptive statistics of our biannual panel for all the relevant variables.

With respect to carbon pricing, Figure 1 panel (a) shows that the number of initiatives implemented in OECD countries has grown sharply in the last decades, going from only 7 initiatives in 2000 to 45 initiatives in 2020. Coverage has also risen steadily. While in 2000 existing OECD initiatives covered together only 0.45% of global emissions, by 2020 the coverage of global emissions from OECD carbon pricing initiatives reached 11%. As shown in Figure 1 panel (b), on a global scale the evolution of the number of initiatives and coverage followed a similar pattern during the period, with 58 initiatives already implemented worldwide by 2020 and an overall coverage of 15% of emissions. It is also worth noting that OECD countries involve a substantial proportion of all the initiatives implemented globally; by considering OECD countries we are accounting for a 78% of initiatives and a 73% of total emissions covered by those initiatives.

Figure 2 further depicts a timeline of the introduction of carbon pricing in OECD countries for the period 2000-2020. As it can be observed, our sample shows substantial variability both in geographical terms and across time, with initiatives of different types –ETS and carbon taxes– implemented in a number of countries and at different points in time. Those initiatives are also heterogeneous in terms of their coverage and price for carbon.

¹⁴They classify exchange rate regimes ranging from 1 to 6, ranging from no separate legal tender to free falling systems or dual markets. The data and a more thorough description can be accessed here: https://carmenreinhart.com/exchange-rate/.

Table 1: Descriptive statistics

		Ν	Mean	SD	Min	Max
	Headline infl. standard deviation (percentage points)	396	0.999	1.214	0.182	15.40
Inflation volatility	Energy infl. standard deviation (percentage points)	396	5.529	3.729	1.048	34.57
	Core infl. standard deviation (percentage points)	374	0.618	0.583	0.101	4.311
	Carbon pricing, any	396	0.672	0.455	0	1
	ETS, any	396	0.595	0.474	0	1
	Carbon Tax, any	396	0.359	0.478	0	1
	Coverage (not overlapping) (%)	396	34.88	31.88	0	130.8
	Coverage ETS (not overlapping) (%)	396	24.73	21.70	0	48.88
	Coverage Carbon Tax (not overlapping) (%)	396	10.57	20.45	0	82.03
	Coverage (fully overlapping) (%)	396	27.97	23.05	0	82.03
Carbon pricing	Coverage ETS (fully overlapping) (%)	396	23.48	21.42	0	48.73
earson prions	Coverage Carbon Tax (fully overlapping) (%)	396	10.39	20.30	0	82.03
	Price $(US\$/tCO_2e)$	396	4.016	9.214	0	83.41
	Price (ETS) $(US\$/tCO_2e)$	396	3.896	4.474	0	14.69
	Price (Carbon Tax) $(US\$/tCO_2e)$	396	3.899	12.63	0	100.8
	Environmental Stringency, 0 (not stringent) to 6 (highest degree)	352	2.546	1.005	0	4.890
	Value-added share of polluting sectors (%)	360	0.141	0.0494	0.0468	0.356
	Post-2015, dummy	396	0.273	0.446	0	1
	GDP per capita (in thousands)	395	39.94	18.24	9.047	113.8
	Population (in thousands)	396	35.17	56.66	0.283	329.5
	Trade openness ($\%$ of GDP)	394	96.45	57.90	19.81	400.6
	Primary exporter (dummy)	396	0.448	0.491	0	1
	Capital flows ($\%$ of GDP)	389	87.74	352.4	3.543	3,365
	Commodity prices: Energy inflation (%)	396	4.843	18.85	-31.97	30.43
	Commodity prices: Agriculture inflation (%)	396	1.286	7.575	-9.476	14.45
Control variables	Commodity prices: Metals & Minerals inf. (%)	396	4.738	16.32	-15.83	31.85
	ExR_Reg No separate legal tender, preannounced peg/band (dummy)	324	0.472	0.498	0	1
	ExR_Reg: Crawling peg/band narrower than $\pm 2\%$ (dummy)	324	0.119	0.318	0	1
	ExR_Reg: Crawling peg/band wider than $\pm 2\%$ (dummy)	324	0.336	0.467	0	1
	ExR_Reg: Free floating (dummy)	324	0.069	0.253	0	1
	Central Bank Independence 0 (lowest) to 1 (highest independence)	252	0.690	0.158	0.179	0.861
	CPI inflation (%)	396	2.503	4.191	-4.990	68.53

N: # of observations; SD: standard deviation



Figure 1: Number of carbon pricing initiatives and coverage

Source: Authors' own elaboration based on World Bank Carbon Pricing Dashboard

Figure 2: Timeline for initiatives of carbon pricing in OECD countries



Source: Authors' own elaboration based on World Bank Carbon Pricing Dashboard

4 Empirical model and estimation strategy

Against the background described above, the econometric model to be estimated can be outlined as follows:

$$\sigma(\pi)_{i,t} = \alpha \sigma(\pi)_{i,t-1} + \beta Carbon \ pricing_{i,t} + \gamma X_{i,t} + \mu_i + \epsilon_{i,t} \tag{1}$$

where $\sigma(\pi)_{i,t}$ is the dependent variable measured, for a country i and for each biennial¹⁵ period t, as the standard deviation of monthly year-on-year inflation computed in windows of 24 months.¹⁶ We considered that inflation volatility could be influenced by its own lag $\sigma(\pi)_{i,t-1}$, showing certain degree of persistence. *Carbon pricing_{i,t}* includes the set of variables that depict the existence and different features of carbon pricing systems, representing our variables of interest (*'Carbon pricing, any', 'ETS, any', 'Carbon tax, any', 'Coverage-fully overlapping', 'Coveragenot overlapping'* and '*Price'*). $X_{i,t}$ include the relevant controls described in the previous section. Finally, μ_i is a country-level time invariant effect that reflects individual heterogeneity, and $\epsilon_{i,t}$ represents an idiosyncratic disturbance that is expected to be independent and identically distributed. α , β , γ , are the coefficients associated with the lag of the dependent variable, and with objective and control variables, respectively. Note that when this past realisation of the dependent variable is also a function of the individual effect, μ_i , the dynamic component is by construction correlated with the error term. To tackle this issue, we employed the System-GMM estimator by Arellano and Bover (1995) and Blundell and Bond (1998).¹⁷

However, if the lag of the dependent variable is not significant the above econometric specification turns into the following static panel:

$$\sigma(\pi)_{i,t} = \beta Carbon \ pricing_{i,t} + \gamma X_{i,t} + \mu_i + \epsilon_{i,t} \tag{2}$$

which could be estimated through the fixed effects or random effects estimators. In the presence

¹⁵Given that our measure of inflation volatility is computed in windows of 24 months, we opted to collapse the rest of the dataset to biennial data (taking averages for both years) to be able to isolate the contemporaneous impact of the objective and control variables. If we had opted for an annual panel, the dependent variable, the 24-month inflation volatility, has been also affected by the previous year evolution of the regressors. As a robustness, we estimated analogous models at annual frequency including contemporaneous and the lagged values of interest and control variables with similar results, which are available upon request.

¹⁶Alternative time windows and additional measures of volatility (see Section 3 for a more thorough description) will be considered as robustness tests in Section 5.5.

¹⁷This estimator incorporates the lagged levels and differences of the dependent variable as instruments. We opt for the two-step GMM alternative, which performs a first-step estimation to retrieve the optimal weighting matrix of instruments that is then used in a second step, using the finite-sample correction derived by Windmeijer (2005) to achieve more efficient estimates.

of correlation between the individual effect and the explanatory variables, the fixed effect estimator should be employed, as the random effect estimator is expected to yield biased and inconsistent estimates of the regression parameters. The Hausman test is usually employed to opt for the fixed or the random effect estimator.¹⁸

5 Results

5.1 Dynamic vs static panel

We first estimated the dynamic panel, considering the potential effect of the persistence of inflation volatility. Table A1 in the Appendix show the estimates of the System-GMM estimator differentiating by type of carbon pricing systems. In none of the specifications considered, the lag of inflation volatility is significant.¹⁹ ²⁰ Therefore, we opt to choose a static panel as our baseline specification.

5.2 Baseline results

Our baseline results are presented in Tables 2 and 3. They correspond to the fixed effect estimator which in this case, after conducting a Hausman test²¹, is preferred over the random effects one to tackle time-invariant unobserved heterogeneity.²²

Table 2 depicts a series of specifications with the objective to test the effects of the existence of any carbon pricing initiative, without further differencing by types. Columns 1 to 5 present some stepwise estimations in which the relevant groups of control variables are sequentially introduced in the specification of the models. In a first step, general socioeconomic variables that account for the size of the country and the level of development are included (column 1). Afterwards, trade openness (column 2), controls for exposure to commodity prices and capital

¹⁸Random effects estimator is preferred under the null hypothesis (zero correlation between the individual effect and the regressors) due to higher efficiency, while under the alternative the fixed effects estimator is at least consistent and, thus, preferred.

¹⁹Although we do not find persistence of inflation volatility at biennial frequency, it does not imply that inflation volatility show persistence at higher frequencies.

²⁰For consistent estimation, in the context of dynamic models the error must be serially uncorrelated. For this purpose, Arellano and Bover (1995) propose a test for the hypothesis of no second order serial correlation of the disturbance. The results of those tests, as well as a Hansen test of over-identifying restrictions reported at the bottom of the table, fail to reject the hypotheses of no serial correlation of order two, and validity of the instruments, respectively. We also report the instrument count.

²¹For all the models, results for the Hausman test reject the null hypothesis that individual effects are uncorrelated with the regressors. Therefore, we present the results of the fixed effects estimator.

 $^{^{22}}$ We also performed a test for normality of the error components in panel-data models proposed by Alejo *et al.* (2015). Our post-estimation results show that the null hypothesis of joint normality of both error terms (individual effect and idiosyncratic component) cannot be rejected.

flows (column 3), variables reflecting common global factors, mainly commodity prices (column 4), and, finally, the domestic level of CPI inflation (column 5) are added.²³ Table 3 presents the same structure as Table 2, but in this case carbon pricing systems are further disaggregated as to consider the ETS and carbon tax types separately.

Standard deviation of the headline CPI inflation							
	(1)	(2)	(3)	(4)	(5)		
Carbon pricing, any	0.610^{***}	0.665^{***}	0.681^{***}	0.696^{***}	0.619^{***}		
	(0.001)	(0.001)	(0.001)	(0.000)	(0.001)		
GDP per capita	-0.064***	-0.056***	-0.070***	-0.056***	-0.048***		
	(0.000)	(0.001)	(0.000)	(0.001)	(0.003)		
Population	-0.042**	-0.049***	-0.050***	-0.042**	-0.029*		
	(0.014)	(0.008)	(0.008)	(0.018)	(0.079)		
Trade openness		-0.004	-0.005	(0.000)	0.000		
D		(0.350)	(0.331)	(0.997)	(0.921)		
Primary exporter			(0.224)	(0.234)	0.179		
Conital flows			(0.287)	(0.250)	(0.334)		
Capital nows			-0.000	-0.000	-0.000		
Commodity prices Energy inflation			(0.438)	(0.900)	(0.855)		
Commonly prices. Energy innation				(0.013)	(0.010^{-10})		
Commodity prices: Agriculture inflation				0.050***	0.065***		
Commonly prices. Agriculture innation				(0.009)	(0.000)		
Commodity prices: Metals & Minerals inf				-0.037***	-0.041***		
Commodity prices. Metals & Minerals ini.				(0,000)	(0.000)		
CPL inflation				(0.000)	0 101***		
					(0.000)		
Constant	4.611***	4.892***	5.426***	4.144***	3.193***		
	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)		
	()	()	()	()	()		
Observations	395	394	387	387	387		
R-squared	0.071	0.075	0.086	0.217	0.314		
Number of countries	36	36	36	36	36		
Hausman test p-value	0.000	0.000	0.000	0.000	0.000		

Table 2: Baseline results: Carbon pricing

Notes: * (**) [***] denotes statistical significance at 10% (5%) [1%]. Robust p-values in parentheses. See main text of the paper for the definition of the variables.

Overall, the existence of a carbon pricing systems is related to increased volatility of inflation (Table 2), which seems to be solely driven by ETS schemes (Table 3). Carbon pricing systems in the form of a carbon tax do not seem to have a significant impact on the volatility of inflation in any of the specifications (Table 3). Regarding the variables of control, higher country size and level of development is related to lower levels of volatility, as expected. Trade openness do not seem to have a significant effect in our sample. On the contrary, being a primary exporter

 $^{^{23}}$ We do not include time fixed effects because they are found highly collinear with some of the control variables, namely, the ones related to global factors. However, in Section 5.5, we show that our results are robust to the inclusion of time fixed effects.

seems to lead to increased volatility of inflation. Global factors, captured by the changes in the different commodity price indices, and also domestic CPI inflation, are significant in all specifications, reflecting their relevance in line with previous findings (Bleaney and Fielding, 2002; Fernández *et al.*, 2017; Parker, 2018; Kamber and Wong, 2020).

Table 3: Baseline results.	ETS	and	carbon	taxes
----------------------------	-----	-----	--------	-------

Standard deviation of the headline CPI inflation								
	(1)	(2)	(3)	(4)	(5)			
	0 705***	0 700***	0.007***	0.750***	0 710***			
E15, any	(0.000)	(0.000)	(0.827^{+++})	(0,000)	(0.000)			
Carbon Torr onr	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)			
Carbon Tax, any	(0.840)	(0.024)	(0.813)	(0.576)	(0.055)			
CDP per capita	-0.075***	-0.067***	-0.082***	-0.067***	-0.058***			
GDI per capita	(0,000)	(0,000)	(0.002)	(0,000)	(0.000)			
Population	-0.033**	-0.038**	-0.036*	-0.030*	-0.018			
1 opulation	(0.048)	(0.032)	(0.050)	(0.084)	(0.266)			
Trade openness	(010-00)	-0.006	-0.006	-0.001	-0.001			
		(0.236)	(0.212)	(0.776)	(0.823)			
Primary exporter		× ,	0.051	0.096	0.047			
			(0.812)	(0.646)	(0.811)			
Capital flows			-0.000	-0.000	-0.000			
			(0.388)	(0.864)	(0.798)			
Commodity prices: Energy inflation				0.014^{***}	0.010^{***}			
				(0.001)	(0.006)			
Commodity prices: Agriculture inflation				0.057***	0.062***			
~				(0.000)	(0.000)			
Commodity prices: Metals & Minerals inf.				-0.037***	-0.041***			
				(0.000)	(0.000)			
CPI inflation					(0.001)			
Constant	1 750***	5 067***	5 501***	1 910***	(0.000)			
Constant	4.758	(0,000)	(0,000)	(0.000)	(0,000)			
	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)			
Observations	395	394	387	387	387			
R-squared	0.081	0.086	0.098	0.223	0.321			
Number of countries	36	36	36	36	36			
Hausman tests	0.000	0.000	0.000	0.000	0.000			

Notes: *(**) [***] denotes statistical significance at 10% (5%) [1%]. Robust p-values in parentheses. See main text of the paper for the definition of the variables.

Table 4 includes additional institutional control variables related to exchange rate regimes and central bank independence. It should be noted that the inclusion of these variables would significantly reduce the sample, since they are available only until 2016 in the case of exchange rate regimes and 2012 for central bank independence. Hence, for the rest of the paper we remain with the specification in which no data is sacrificed and main controls are included (column 5 in Tables 2 and 3). However, we should stress that our main results are robust to the inclusion of those additional controls.

Standard deviation of the headline CPI inflation				
	(1)	(2)	(3)	(4)
Carbon pricing, any	0.492^{***}	0.656^{**}		
ETS, any	(0.001)	(0.011)	0.479^{***}	0.854^{***}
Carbon Tax, any			(0.002) 0.145	(0.004) -0.193
GDP per capita	-0.003	-0.052	(0.393) -0.012	(0.645) - 0.082^*
Population	(0.848) 0.000 (0.984)	(0.201) - 0.094^{***} (0.008)	(0.467) 0.010 (0.487)	(0.058) - 0.086^{**} (0.014)
Trade openness	-0.002	-0.003	-0.002	(0.014) -0.005 (0.560)
Primary exporter	(0.349) 0.214 (0.240)	(0.703) 0.806^{**} (0.027)	(0.500) 0.081 (0.660)	(0.303) 0.715^{*} (0.051)
Capital flows	(0.240) 0.000 (0.070)	(0.027) 0.000 (0.645)	(0.009) 0.000 (0.022)	(0.031) 0.000 (0.621)
Commodity prices: Energy inflation	(0.979) 0.014^{***}	(0.645) 0.008	(0.923) 0.015^{***}	(0.631) 0.010 (0.228)
Commodity prices: Agriculture inflation	(0.000) 0.058***	(0.444) 0.069^{***}	(0.000) 0.056***	(0.338) 0.066***
Commodity prices: Metals & Minerals inf.	(0.000) -0.033***	(0.000) - 0.042^{***}	(0.000) -0.033***	(0.000) - 0.043^{***}
CPI inflation	(0.000) -0.059***	(0.000) 0.096***	(0.000) -0.059***	(0.000) 0.097^{***}
${\rm ExR_Reg}$ No separate legal tender, preannounced peg/band	(0.000) -27.838*** (0.000)	(0.000)	(0.000) -27.830*** (0.000)	(0.000)
ExR_Reg: Crawling peg/band narrower than $\pm 2\%$	-26.938^{***}		-26.905^{***}	
ExR_Reg: Crawling peg/band wider than $\pm 2\%$	-26.769^{***}		-26.794^{***}	
ExR_Reg: Free floating	-26.686***		-26.417***	
Central Bank Independence	(0.000)	-0.132 (0.954)	(0.000)	$\begin{array}{c} 0.417 \\ (0.856) \end{array}$
Observations	318	246	318	246
R-squared	0.718	0.321	0.719	0.332
Number of countries	36	36	36	36

Table 4: Robustness to the introduction of other control variables

Notes: * (**) [***] denotes statistical significance at 10% (5%) [1%]. Robust p-values in parentheses. See main text of the paper for the definition of the variables.

5.3Channels. Effects on the volatility of energy and core inflation

In second place, we investigate the channels by which carbon pricing systems affect inflation volatility, analysing the differential impact on the energy CPI component and the degree of passthrough to the volatility of core inflation. For this purpose, in Table 5 the dependent variables are headline, energy and core inflation volatility, respectively. We consider the overall effect of having implemented any carbon pricing initiative (columns 1, 2 and 3) and the effect of ETS and carbon taxes separately (columns 4, 5 and 6). Control variables from the baseline specification are also included, in a similar vein as in column 5 in the Tables 2 and 3.

The estimates reveal that carbon pricing systems show the highest impact on the volatility

	(1) Sd.Headline	(2) Sd.Energy	(3) Sd.Core	(4) Sd.Headline	(5) Sd.Energy	(6) Sd.Core
Carbon pricing, any	0.619^{***} (0.001)	1.014^{*}	-0.040			
ETS, any	(0001)	(0.001)	(0.000)	0.710***	1.122*	-0.058
Carbon Tax, any				(0.000) 0.055 (0.778)	(0.052) 0.599 (0.314)	(0.571) -0.096 (0.375)
Observations	387	387	367	387	387	367
R-squared	0.314	0.360	0.161	0.321	0.365	0.164
Number of countries Control variables included	36 YES	36 YES	34 YES	36 YES	$\frac{36}{\text{YES}}$	34 YES

Table 5: Energy and core inflation

Notes: * (**) [***] denotes statistical significance at 10% (5%) [1%]. Robust p-values in parentheses. Control variables: GDP per capita, Population, Trade openness, Primary exporter, Capital flows, Commodity prices: Energy inflation, agriculture inflation and Metals & Minerals inflation, CPI inflation. See main text of the paper for the definition of the variables.

of inflation for the case of energy CPI, which should be expected given that energy related goods are the ones that are targeted directly or indirectly by carbon pricing initiatives. On the contrary, we do not find a significant impact of carbon pricing initiatives on the volatility of core inflation. These results are in line with previous findings that the introduction of carbon pricing schemes seems to impact inflation through the energy component, and do not tend to affect core inflation, which seems consistent with the fact that by increasing energy prices, carbon pricing change relative prices, but do not necessarily lead to increases in the price of the rest of the basket of goods and services (Konradt and Weder di Mauro, 2021; Mckibbin *et al.*, 2021; Moessner, 2022). There is also some evidence that the indirect effects derived from the increase in the cost of producing goods and services due to higher costs of energy and intermediate inputs, and its pass-through to prices, may be not significant. In fact, Álvarez *et al.* (2011) find, for Spain and the euro area, that the indirect effect of a shock in oil prices tend to transmit at a substantially lower speed to consumer prices, and that those effects tend indeed to be limited as opposed to the direct effects. Consistent with those results, the effect on the volatility of core inflation is found insignificant, at least at the carbon price levels and coverage from our sample.

Regarding the type of carbon pricing initiative affecting inflation volatility, again, it seems that schemes based on the ETS are the ones that spur the increased volatility pattern, but only for headline and energy inflation volatility (columns 5 and 6), while core inflation does not seem to be affected. The effect of ETS on the volatility of energy inflation is also larger than the one on headline inflation (columns 4 and 5). We cannot find a significant effect on carbon taxes on the volatility of any constituent of the CPI.

5.4 GHG emissions coverage and effective carbon price

As we mentioned in Section 2, the effect of carbon pricing systems on inflation volatility may depend on the amount of the GHG emissions covered by the carbon pricing initiatives in each country and the effective carbon price paid in this jurisdiction. Table 6 summarises the effect of coverage and the level of carbon prices on inflation volatility. These specifications also considered the full set of controls although they are not reported for convenience.

Standard deviation of the headline CPI inflation GHG coverage Effective carbon price (1)(2)(4)(6)(3)(5)0.013*** Coverage (not overlapping) (0.000)0.017*** Coverage ETS (not overlapping) (0.000)Coverage Carbon Tax (not overlapping) 0.008 (0.138)Coverage (fully overlapping) 0.015*** (0.002)0.016*** Coverage ETS (fully overlapping) (0.001)Coverage Carbon Tax (fully overlapping) 0.007 (0.212)0.010 Price (0.466)Price (ETS) 0.045^{*} (0.026)Price (Carbon Tax) 0.001 (0.891)Observations 387 387387387 387387 0.3220.2910.296R-squared 0.3190.3220.313Number of countries 363636 36 3636YES YES YES YES YES Controls included YES

Table 6: Results by GHG emissions coverage and effective carbon price

Notes: * (**) [***] denotes statistical significance at 10% (5%) [1%]. Robust p-values in parentheses. Control variables: GDP per capita, Population, Trade openness, Primary exporter, Capital flows, Commodity prices: Energy inflation, agriculture inflation and Metals & Minerals inflation, CPI inflation. See main text of the paper for the definition of the variables.

As shown in columns 1 to 4 of Table 6, the GHG emissions coverage of the carbon pricing initiatives significantly affects the volatility of inflation. That is, the higher the level of coverage, the higher the volatility of headline inflation. In addition, the result is robust to the use of the two alternative measures of coverage ('*Coverage-not overlapping*' and '*Coverage-fully overlapping*') described in Section 3. When considering the effect of the coverage disaggregating by type of

carbon price systems, once again, the effect appears to be driven by the ETS schemes, being not significant in the case of carbon taxes (columns 2 and 4). Regarding the effective carbon price level, our results point to higher carbon prices increasing the volatility of headline inflation (column 5), an effect again explained by the ETS (column 6).

5.5 Additional robustness tests

Finally, in this section we present further evidence of our results being robust to the use of different specifications, subsamples and additional controls.

First, Table A2 in the appendix test our hypotheses described in Section 2 that the effect on carbon pricing systems on inflation could depend on a country's level of specialization in high polluting-sectors and industries, and the existence of a structural break after the renewed surge in environmental policies following the Paris Agreement in 2015, finding no evidence of any of the two. Also, our results are proved robust to the inclusion of the OECD index of environmental stringency, in an attempt to control for the fact that our variables of carbon pricing could be capturing the effects of changes in each countries environmental stringency not necessarily related. A more stringent environmental regulation does not seem nevertheless to affect the volatility of inflation.

Second, we test our specification against the use of different time windows for the computation of volatility and alternative measures, as described in Section 2. As shown in Tables A3 and A4, our main results are robust to the use of windows of 12 and 36 months (Table A3), and the log of one plus the standard deviation of inflation (Table A4).

Third, given that the classification of advanced economies differs among organizations, we restrict our sample to those countries that are considered as advanced economies according to the International Monetary Fund, using the classification of its World Economic Outlook. As depicted in Table A5, in this restricted subsample, carbon pricing systems keep showing a significant positive effect on inflation volatility, driven by the systems of the ETS type.

Finally, we add a specification in which time dummies are included, at the expense of removing those variables collinear with them, the global factors captured by the international commodity prices, showing that main results hold under this alternative modelling choice (Table A6).

5.6 Robustness to the use of local projections

One of the studies on the impact of carbon pricing initiatives on the level of inflation (Konradt and Weder di Mauro, 2021)²⁴ has also considered the dynamic responses by using panel local projections following Jordà (2005). In this section, we apply this methodology to inflation volatility to check to what extent our previous results are robust to alternative empirical approaches.

We choose the following specification to estimate the dynamic response of the volatility of CPI inflation at horizon h to a carbon price shock.

$$\sigma(\pi)_{i,t+h} = \Theta^h Carbon \, price_{i,t} + \beta^h(L) \, Carbon \, price_{i,t-1} + \delta^h(L)\sigma(\pi)_{i,t-1} + \gamma^h X_{i,t} + \mu^h_i + \epsilon^h_{i,t} \quad (3)$$

where $\sigma(\pi)_{i,t}$ is the standard deviation of monthly year-on-year inflation for a country i and for the year t, computed in windows of 24 months. We consider the standard deviation of headline, energy and core inflation. *Carbon price_{i,t}* is the real effective carbon price, for which we distinguish among the associated to any carbon pricing initiative, to ETS and to carbon taxes. $X_{i,t}$ includes GDP per capita, to control for the level of development, and the growth rate of energy commodity prices, for global factors. μ_i^h is a country fixed effect and $\epsilon_{i,t}^h$ represents an error term. Under the identifying assumption that a component of the effective carbon price is not predictable by past macroeconomic controls, Θ^h can be interpreted as the effect of an unexpected change in the carbon price in year t on inflation volatility in h years. Finally, $\beta^h(L)$ represents a lag polynomial, reflecting the use of the four latest lags to control for persistence of the carbon pricing and inflation volatility.

The annual impulse responses shown in Figure 3 are constructed based on the fixed effects estimates of Θ^h coefficients at each horizon. Following Metcalf and Stock (2020) and Konradt and Weder di Mauro (2021), the shock consists in a one-time permanent increase in the effective carbon price by 40 USD applied to 30% of each economy's GHG emissions, equivalent to an increase of the effective carbon price of 12 USD. The 95% confidence bands reported rely on the respective estimated standard errors clustered by country and heteroskedasticity robust.

The main conclusions extracted from this analysis could be summarized as follows. First, a rise in effective carbon prices is associated to an increase of the volatility of headline inflation. This is specially clear for ETS prices whereas the impact of carbon taxes rates is only found

²⁴Konradt and Weder di Mauro (2021) adapt the specification employed by Metcalf and Stock (2020) to assess the impact of carbon taxes on economic growth.

to marginally reduce volatility for one of the years. Second, effective carbon prices push up the volatility of energy inflation, although only through ETS prices since the dynamic impact of carbon tax rates is not significant in the horizon. Third, none of the prices of any of these initiatives seems to affect the volatility of core inflation. All in all, the baseline results based on

a static panel data analysis hold roughly robust to the dynamic local projections approach.

Figure 3: Impulse responses of headline, energy and core inflation volatility to shocks in carbon prices, ETS prices and carbon tax rates



Notes: The blue solid lines describe the dynamic impulse responses to a 40 USD carbon tax with 30% emission coverage. The grey area delimits 95% confidence bands. All responses are calculated by local projections including country fixed effects and controlling for GDP per capita and the growth rate of energy commodity prices. The variables enter the regression with four lags. Standard errors are heteroskedasticity robust and clustered by country.

6 Conclusions

In the following years, climate change and related mitigation policies are expected to affect potential growth, business cycle and inflation dynamics. This may reshape the formulation of economic policy, and has prompted the urge from policy makers, including central banks, to understand their economic consequences.

Carbon pricing initiatives, aimed at increasing the relative prices of GHG intensive good and services, could not only push up inflation but also affect the volatility of headline inflation. This paper assesses empirically the effects of carbon pricing on inflation volatility disaggregating by the two main types of initiatives: ETS and carbon taxes. To our knowledge, while some other works have dealt with their impact on the level of inflation, this is the first paper that addresses the effects on carbon pricing schemes on the volatility of that inflation.

Our main findings could be summarised as follows. First, this work finds strong evidence that carbon pricing schemes are associated with larger CPI headline inflation volatility. Importantly, this effect is found to be driven only by systems of the ETS type, while no impact is found for the case of carbon taxes. Second, we find that inflation volatility seems to be fed only through the energy component. That is to say, the effect of carbon pricing on the volatility of inflation stemming from ETS systems seems circumscribed to energy inflation, and no significant impact is found for the volatility of the core inflation. Third, only for ETS systems, both the increase in the underlying price of emissions and the expansion of the activities covered by carbon pricing initiatives are associated with greater inflation volatility. Finally, our results hold robust to the inclusion of multiple variables of control, different empirical specifications and subsamples of countries.

Altogether, our results seem consistent with the fact that carbon pricing initiatives of the ETS type tend to lead to more volatile carbon prices (and carbon price inflation) than the more stable carbon taxes, and that volatility is fed into consumer prices, particularly through the energy component. It also confirms the results of previous studies that carbon pricing initiatives, by changing relative prices and increasing the price of fossil fuels, led to an effect on inflation by affecting directly the energy component, but they do not lead to significant indirect effects in the goods and services part of the core CPI, at least at the carbon price levels and coverage from our sample. Our results find additional evidence that those effects also ultimately translate to higher volatility of the headline inflation through the energy component.

These results have also important implications for monetary policy. Since existing empirical literature has only found that carbon pricing schemes are generally associated to a transitory effect on the level of inflation, credible monetary policy authorities will generally 'look through' this effect as it shares the features of a negative supply-side shock. However, as ETS is found to affect inflation volatility, those relative price shocks further complicate the conduct of monetary policy. Since both the coverage of carbon pricing schemes and the price for carbon should

increase in order to achieve the ambitious climate goals of the Paris Agreement, larger inflation volatility could be expected, especially in those jurisdictions in which ETS play a leading role as a mitigation policy. This could affect the communication and the conduct of the monetary policy, and should therefore be monitored closely by central banks. In addition, our results also point to the need to consider the introduction of corrective mechanisms to reduce the volatility of those systems.

Finally, further research is needed to better understand how mitigation policies affect other relevant dimensions of the inflation process, beyond its level and volatility. Particularly for the conduct of monetary policy, it seems of great interest to address the implications of carbon pricing initiatives on inflation persistence, inflation expectations or inflation forecast errors.

Appendix. Figures and Tables



Figure A1: Standard deviation of CPI inflation

Source: Authors' own elaboration based on OECD Data

	(1)	(2)
	Any carbon pricing	ETS and carbon taxes
L. inflation volatility	0.032	0.032
	(0.671)	(0.648)
Carbon pricing, any	-0.338	
	(0.478)	
ETS, any		-0.276
		(0.507)
Carbon Tax, any		0.095
		(0.809)
GDP per capita	0.003	-0.000
	(0.694)	(0.997)
Population	0.002	0.002
	(0.194)	(0.244)
Trade openness	0.007^{**}	0.007^{**}
	(0.020)	(0.037)
Primary exporter	0.194	0.156
	(0.254)	(0.324)
Capital flows	-0.001**	-0.001***
	(0.021)	(0.008)
Commodity prices: Energy inflation	0.011^{***}	0.010^{***}
	(0.005)	(0.004)
Commodity prices: Agriculture inflation	0.074^{***}	0.076^{***}
	(0.000)	(0.000)
Commodity prices: Metals & Minerals inf.	-0.049***	-0.050***
	(0.000)	(0.000)
CPI inflation	0.160^{***}	0.158^{***}
	(0.000)	(0.000)
Observations	355	355
Number of countries	36	36
Hansen test p-value	0.215	0.197
Arellano-Bond test $AR(1)$ p-value	0.00571	0.00382
Arellano-Bond test $AR(2)$ p-value	0.504	0.461
Instrument count	34	34

Table A1: System-GMM estimates of the dynamic panel

Notes: * (**) [***] denotes statistical significance at 10% (5%) [1%]. Robust p-values in parentheses. See main text of the paper for the definition of the variables.

Standard deviation of the headline CPI inflation						
	(1)	(2)	(3)	(4)	(5)	(6)
Carbon pricing, any	0.619^{***}		1.396^{**}		0.583^{***}	
ETS, any	(0.001)	0.777***	(0.010)	1.069**		0.837***
Carbon Tax, any		$(0.000) \\ -0.058 \\ (0.797)$		(0.030) -0.015 (0.981)		(0.000) -0.021 (0.9160)
Post-2015, dummy	0.345	0.490*				
Carbon pricing, any * Post-2015, dummy	(0.364) -0.246 (0.525)	(0.099)				
ETS, any * Post-2015, dummy	· /	-0.360				
Carbon Tax, any * Post-2015, dummy		(0.233) -0.005 (0.985)				
Value-added share of polluting sectors (20%)		· · ·	3.597	0.457		
Carbon pricing, any *Value-added share polluting sectors			$ \begin{array}{c} (0.453) \\ -5.425 \\ (0.178) \end{array} $	(0.923)		
ETS, any * Value-added share polluting sectors				-2.556		
Carbon tax, any * Value-added share polluting sectors				(0.432) 0.865 (0.851)		
Environmental Stringency					-0.040	-0.167
					(0.729)	(0.186)
Observations	387	387	354	354	343	343
R-squared	0.316	0.327	0.323	0.325	0.335	0.349
Control variables included	30 YES	30 YES	YES	30 YES	YES	$_{\rm YES}^{32}$

Table A2: Results considering period post Paris Agreement, the share in value added of polluting sectors and environmental stringency

Notes: * (**) [***] denotes statistical significance at 10% (5%) [1%]. Robust p-values in parentheses. Control variables: GDP per capita, Population, Trade openness, Prima exporter, Capital flows, Commodity prices: Energy inflation, agriculture inflation and Metals & Minerals inflation, CPI inflation. See main text of the paper for the definition of the variables.

Standard deviation of the headline CPI inflation								
	(1) 12 months	(2) 24 months	(3) 36 months	(4)12 months	(5) 24 months	(6) 36 months		
Carbon pricing, any	0.239^{***} (0.000	0.619^{***} (0.001)	0.573^{***} (0.006)					
ETS, any	,	· · · ·	· · · ·	0.305^{***}	0.710^{***}	0.618^{***}		
				(0.000)	(0.000)	(0.009)		
Carbon Tax, any				-0.021	0.055	0.304		
				(0.764)	(0.778)	(0.159)		
Observations	716	387	249	716	387	249		
R-squared	0.456	0.314	0.414	0.464	0.321	0.421		
Number of countries	36	36	36	36	36	36		
Control variables included	YES	YES	YES	YES	YES	YES		

Notes: * (**) [***] denotes statistical significance at 10% (5%) [1%]. Robust p-values in parentheses. Control variables: GDP per capita, Population, Trade openness, Primary exporter, Capital flows, Commodity prices: Energy inflation, agriculture inflation and Metals & Minerals inflation, CPI inflation. See main text of the paper for the definition of the variables.

Log(1+standard deviation headline CPI inflation)						
	(1)	(2)				
Carbon pricing, any	0.187^{***} (0.000)					
ETS, any		0.229***				
Carbon Tax, any		$(0.000) \\ -0.008 \\ (0.886)$				
Observations	387	387				
R-squared	0.361	0.372				
Number of countries	36	36				
Controls included	YES	YES				

Table A4: Robustness checks. Volatility measure in logarithms

Notes: * (**) [***] denotes statistical significance at 10% (5%) [1%]. Robust p-values in parentheses. Control variables: GDP per capita, Population, Trade openness, Primary exporter, Capital flows, Commodity prices: Energy inflation, agriculture inflation and Metals & Minerals inflation, CPI inflation. See main text of the paper for the definition of the variables.

	(1)	(2)
Carbon pricing, any	0.524^{***} (0.000)	
ETS, any		0.640^{***} (0.000)
Carbon Tax, any		(0.000) (0.009) (0.946)
Observations	299	299
R-squared	0.411	0.431
Number of countries	28	28
Control variables included	YES	YES

Table A5: Robustness checks. Alternative advanced economies grouping

Standard deviation of the headline CPI inflation

Notes: * (**) [***] denotes statistical significance at 10% (5%) [1%]. Robust p-values in parentheses. Control variables: GDP per capita, Population, Trade openness, Primary exporter, Capital flows, Commodity prices: Energy inflation, agriculture inflation and Metals & Minerals inflation, CPI inflation. See main text of the paper for the definition of the variables.

Standard deviation of the headline CPI inflation							
	Baseline results		GHG coverage		Effective carbon price		
	(1)	(2)	(3)	(4)	(5)	(6)	
Carbon pricing, any	0.507^{**} (0.014)						
ETS, any	· · /	0.694^{***} (0.005)					
Carbon Tax, any		0.058 (0.774)					
Coverage (fully overlapping)		. ,	0.011^{*} (0.055)				
Coverage ETS (fully overlapping)			· · · ·	0.014^{**} (0.026)			
Coverage Carbon Tax (fully overlapping)				0.008 (0.197)			
Price				. ,	$0.005 \\ (0.743)$		
Price (ETS)						0.060^{**} (0.024)	
Price (Carbon tax)						-0.002 (0.866)	
Observations	387	387	387	387	387	387	
R-squared	0.340	0.344	0.337	0.344	0.328	0.335	
Number of countries	36	36	36	36	36	36	
Control variables included	YES	YES	YES	YES	YES	YES	
Time FE	YES	YES	YES	YES	YES	YES	

Table A6: Robustness	checks.	Inclusion	of	time	dummies
----------------------	---------	-----------	----	-----------------------	---------

Notes: * (**) [***] denotes statistical significance at 10% (5%) [1%]. Robust p-values in parentheses. Control variables: GDP per capita, Population, Trade openness, Primary exporter, Capital flows, Commodity prices: Energy inflation, agriculture inflation and Metals & Minerals inflation, CPI inflation. Biennial time fixed effects included. Results including coverage non-overlapping are qualitatively similar and available upon request. See main text of the paper for the definition of the variables.

References

- Aghion, P., P. Bacchetta and A. Banerjee (2004). "Financial development and the instability of open economies", *Journal of Monetary Economics*, 51, pp. 1077-1106.
- Alejo, J., A. Galvao, G. Montes-Rojas and W. Sosa-Escudero (2015). "Tests for normality in linear panel-data models", *Stata Journal*, 15, pp. 822-832.
- Alfaro, L. (2005). "Inflation, openness, and exchange-rate regimes: The quest for short-term commitment", *Journal of Development Economics*, 77, pp. 229-249.
- Álvarez, L. J., S. Hurtado, I. Sánchez and C. Thomas (2011). "The impact of oil price changes on Spanish and euro area consumer price inflation", *Economic Modelling*, 28, pp. 422-431.
- Andersson, J. J. (2019). "Carbon taxes and CO₂ emissions: Sweden as a case study", *American Economic Journal: Economic Policy*, 11, pp. 1-30.
- Arellano, M., and O. Bover (1995). "Another look at the instrumental variable estimation of error-components models", *Journal of Econometrics*, 68, pp. 29-51.
- Balatti, M. (2020). *Inflation volatility in small and large advanced open economies*, Working Paper Series 2448, European Central Bank.
- Bernanke, B. S., and M. Gertler (2001). "Should central banks respond to movements in asset prices?", *The American Economic Review*, 91, pp. 253-257.
- Blanchard, O., E. Cerutti and L. H. Summers (2015). *Inflation and Activity: Two Explorations and Their Monetary Policy Implications*, Working Paper Series WP15-19, Peterson Institute for International Economics.
- Bleaney, M., and D. Fielding (2002). "Exchange rate regimes, inflation and output volatility in developing countries", *Journal of Development Economics*, 68, pp. 233-245.
- Blundell, R., and S. Bond (1998). "Initial conditions and moment restrictions in dynamic panel data models", *Journal of Econometrics*, 87, pp. 115-143.
- Bowdler, C., and A. Malik (2005). *Openness and inflation volatility: Cross-country evidence,* Economics Papers 2005-W14, Economics Group, Nuffield College, University of Oxford.
- Bowdler, C., and A. Malik (2017). "Openness and inflation volatility: Panel data evidence", *The North American Journal of Economics and Finance*, 41, pp. 57-69.
- Cukierman, A., S. B. Webb and B. Neyapti (1992). "Measuring the independence of central banks and its effect on policy outcomes", *The World Bank Economic Review*, 6, pp. 353-398.

- Elkins, P., and T. Baker (2001). "Carbon taxes and carbon emissions trading", *Journal of Economic Surveys*, 15, pp. 325-376.
- Estrada, Á., and D. Santabárbara (2021). *Recycling carbon tax revenues in Spain. Environmental and economic assessment of selected green reforms,* Working Papers 2119, Banco de España.
- Fernández, A., S. Schmitt-Grohé and M. Uribe (2017). "World shocks, world prices, and business cycles: An empirical investigation", *Journal of International Economics*, 108, S2-S14, 39th Annual NBER International Seminar on Macroeconomics.
- Forbes, K. (2019). *Inflation dynamics: Dead, dormant, or determined abroad?*, Working Paper 26496, National Bureau of Economic Research.
- Garriga, A. C. (2016). "Central bank independence in the world: A new data set", *International Interactions.*
- Goulder, L. H., and A. R. Schein (2013). "Carbon taxes versus cap and trade: A critical review", *Climate Change Economics (CCE)*, 04, pp. 1-28.
- Ha, J., A. Kose and F. Ohnsorge (2019). *Inflation in emerging and developing economies: evolution, drivers and policies,* World Bank, p. 490.
- Hepburn, C., J. Stiglitz and N. Stern (2020). "Carbon Pricing", special issue, *European Economic Review*, 127, 103440.
- Ilzetzki, E., C. M. Reinhart and K. S. Rogoff (2019). "Exchange Arrangements Entering the Twenty-First Century: Which Anchor will Hold?", *The Quarterly Journal of Economics*, 134, pp. 599-646.
- IPCC (2022). Climate change 2022: impacts, adaptation and vulnerability, 6.
- Jordà, Ó. (2005). "Estimation and inference of impulse responses by local projections", *American Economic Review*, 95, pp. 161-182.
- Kamber, G., and B. Wong (2020). "Global factors and trend inflation", *Journal of International Economics*, 122, 103265.
- Känzig, D. R. (2021). The unequal economic consequences of carbon pricing.
- Klomp, J., and J. de Haan (2010). "Central bank independence and inflation revisited", *Public Choice*, 144, pp. 445-457.
- Konradt, M., and B. Weder di Mauro (2021). *Carbon Taxation and Greenflation: Evidence from Europe and Canada*, IHEID Working Papers 17-2021, Economics Section, The Graduate Institute of International Studies.

- Lane, P. R. (1997). "Inflation in open economies", Journal of International Economics, 42, pp. 327-347.
- Martin, R., M. Muûls, L. B. de Preux and U. J. Wagner (2014). "Industry compensation under relocation risk: A firm-level analysis of the EU Emissions Trading Scheme", *American Economic Review*, 104, pp. 2482-2508.
- Martin, R., M. Muûls and U. J. Wagner (2016). "The impact of the European Union Emissions Trading Scheme on regulated firms: What is the evidence after ten years?", *Review of Environmental Economics and Policy*, 10, pp. 129-148.
- Mckibbin, W., M. Konradt and B. Weder di Mauro (2021). "Climate policies and monetary policies in the euro area", *ECB Sintra Forum* 2021, pp. 1-41.
- McKibbin, W., A. Morris and P. Wilcoxen (2014). *The economic consequences of delay in US climate policy*, CAMA Working Paper, p. 35.
- Metcalf, G. E. (2021). "Carbon taxes in theory and practice", Annual Review of Resource Economics, 13, pp. 245-265.
- Metcalf, G. E., and J. H. Stock (2020). "Measuring the macroeconomic impact of carbon taxes", *AEA Papers and Proceedings*, 110, pp. 101-06.
- Moessner, R. (2022). Effects of carbon princing on inflation, CESifo Working Paper, p. 20.
- Mundaca, G., J. Strand and I. R. Young (2021). "Carbon pricing of international transport fuels: Impacts on carbon emissions and trade activity", *Journal of Environmental Economics and Management*, 110, 102517.
- Neely, C. J., and D. E. Rapach (2011). "International comovements in inflation rates and country characteristics", *Journal of International Money and Finance*, 30, pp. 1471-1490.
- NGFS (2021). Annual report 2020, pp. 1-12.
- Parker, M. (2018). "How global is 'global inflation'?", Journal of Macroeconomics, 58, pp. 174-197.
- Pigou, A. (1932). "The Economics of Welfare. By A. C. Pigou. Fourth Edition", *The Economic Journal*, 43, pp. 329-330.
- Reinhart, C. M., and K. S. Rogoff (2004). "The Modern History of Exchange Rate Arrangements: A Reinterpretation", *The Quarterly Journal of Economics*, 119, pp. 1-48.
- Romer, D. (1993). "Openness and inflation: theory and evidence", *The Quarterly Journal of Economics*, 108, pp. 869-903.
- Schnabel, I. (2021). "A new strategy for a changing world", Speech at the virtual Financial Statements series hosted by the Peterson Institute for International Economics.
- Schnabel, I. (2022). "Looking through higher energy prices? Monetary policy and the Green Transition", Speech at a panel on "Climate and the financial system" at the American Financial Association 2022 Virtual Annual Meeting.

- Shapiro, A., and G. E. Metcalf (2021). *The macroeconomic effects of a carbon tax to meet the U.S. Paris agreement target: The role of firm creation and technology adoption,* Working Paper 28795, National Bureau of Economic Research.
- Venmans, F., J. Ellis and D. Nachtigall (2020). "Carbon pricing and competitiveness: are they at odds?", *Climate Policy*, 20, pp. 1070-1091.
- Windmeijer, F. (2005). "A finite sample correction for the variance of linear efficient two-step GMM estimators", *Journal of Econometrics*, 126, pp. 25-51.

World Bank (2021). State and trends of carbon pricing 2021, pp. 1-87.

Yamazaki, A. (2017). "Jobs and climate policy: Evidence from British Columbia's revenueneutral carbon tax", *Journal of Environmental Economics and Management*, 83, pp. 197-216.

BANCO DE ESPAÑA PUBLICATIONS

WORKING PAPERS

- 2111 YUNUS AKSOY, HENRIQUE S. BASSO and CAROLYN ST AUBYN: Time Variation in Lifecycle Consumption and Income.
- 2112 JENNIFER PEÑA and ELVIRA PRADES: Price setting in Chile: micro evidence from consumer on-line prices during the social outbreak and Covid-19.
- 2113 NEZIH GUNER, YULIYA A. KULIKOVA and ARNAU VALLADARES-ESTEBAN: Does the added worker effect matter?
- 2114 RODOLFO G. CAMPOS and JACOPO TIMINI: Unequal trade, unequal gains: the heterogeneous impact of MERCOSUR.
- 2115 JAVIER QUINTANA: Import competition, regional divergence, and the rise of the skilled city.
- 2116 PATRICK MACNAMARA, MYROSLAV PIDKUYKO and RAFFAELE ROSSI: Marginal Tax Changes with Risky Investment.
- 2117 RODOLFO G. CAMPOS, JACOPO TIMINI and ELENA VIDAL: Structural gravity and trade agreements: does the measurement of domestic trade matter?
- 2118 ALESSANDRO MELCARNE, JUAN S. MORA-SANGUINETTI and ROK SPRUK: Democracy, technocracy and economic growth: evidence from 20 century Spain.
- 2119 ÁNGEL ESTRADA and DANIEL SANTABÁRBARA: Recycling carbon tax revenues in Spain. Environmental and economic assessment of selected green reforms.
- 2120 ALEJANDRO FERNÁNDEZ-CEREZO, BEATRIZ GONZÁLEZ, MARIO IZQUIERDO and ENRIQUE MORAL-BENITO: Firm-level heterogeneity in the impact of the COVID-19 pandemic.
- 2121 EDUARDO GUTIÉRREZ and CÉSAR MARTÍN MACHUCA: The effect of tariffs on Spanish goods exports.
- 2122 JACOPO TIMINI: Revisiting the 'Cobden-Chevalier network' trade and welfare effects.
- 2123 ALEJANDRO BUESA, JAVIER J. PÉREZ and DANIEL SANTABÁRBARA: Awareness of pandemics and the impact of COVID-19.
- 2124 ALICIA AGUILAR and DIEGO TORRES: The impact of COVID-19 on analysts' sentiment about the banking sector.
- 2125 SILVIA ALBRIZIO, IVÁN KATARYNIUK, LUIS MOLINA and JAN SCHÄFER: ECB euro liquidity lines.
- 2126 ANTHONY BALD, ERIC CHYN, JUSTINE HASTINGS and MARGARITA MACHELETT: The causal impact of removing children from abusive and neglectful homes.
- 2127 IRMA ALONSO, PEDRO SERRANO and ANTONI VAELLO-SEBASTIÀ: The impact of heterogeneous unconventional monetary policies on the expectations of market crashes.
- 2128 MARÍA T. GONZÁLEZ-PÉREZ: Lessons from estimating the average option-implied volatility term structure for the Spanish banking sector.
- 2129 SIMÓN A. RELLA, YULIYA A. KULIKOVA, EMMANOUIL T. DERMITZAKIS and FYODOR A. KONDRASHOV: Rates of SARS-COV-2 transmission and vaccination impact the fate of vaccine-resistant strains.
- 2130 MATÍAS LAMAS and DAVID MARTÍNEZ-MIERA: Sectorial holdings and stock prices: the household-bank nexus.
- 2131 ALBERT BANAL-ESTAÑOL, ENRIQUE BENITO, DMITRY KHAMETSHIN and JIANXING WEI: Asset encumbrance and bank risk: theory and first evidence from public disclosures in Europe.
- 2132 ISABEL ARGIMÓN and MARÍA RODRÍGUEZ-MORENO: Business complexity and geographic expansion in banking.
- 2133 LUIS GUIROLA: Does political polarization affect economic expectations?: Evidence from three decades of cabinet shifts in Europe.
- 2134 CHRISTIANE BAUMEISTER, DANILO LEIVA-LEÓN and ERIC SIMS: Tracking weekly state-level economic conditions.
- 2135 SERGI BASCO, DAVID LÓPEZ-RODRÍGUEZ and ENRIQUE MORAL-BENITO: House prices and misallocation: The impact of the collateral channel on productivity.
- 2136 MANUEL ARELLANO, STÉPHANE BONHOMME, MICOLE DE VERA, LAURA HOSPIDO and SIQI WEI: Income risk inequality: Evidence from Spanish administrative records.
- 2137 ANGELA ABBATE and DOMINIK THALER: Optimal monetary policy with the risk-taking channel.
- 2138 MARTA BANBURA, DANILO LEIVA-LEÓN and JAN-OLIVER MENZ: Do inflation expectations improve model-based inflation forecasts?
- 2139 MÁXIMO CAMACHO, MARÍA DOLORES GADEA and ANA GÓMEZ LOSCOS: An automatic algorithm to date the reference cycle of the Spanish economy.
- 2140 EDUARDO GUTIÉRREZ, AITOR LACUESTA and CÉSAR MARTÍN MACHUCA: Brexit: Trade diversion due to trade policy uncertainty.
- 2141 JULIO A. CREGO and JULIO GÁLVEZ: Cyclical dependence in market neutral hedge funds.
- 2142 HERVÉ LE BIHAN, MAGALI MARX and JULIEN MATHERON: Inflation tolerance ranges in the new keynesian model.
- 2143 DIEGO COMIN, JAVIER QUINTANA, TOM SCHMITZ and ANTONELLA TRIGARI: Measuring TFP: the role of profits, adjustment costs, and capacity utilization.

- 2144 ROBERTO PASCUAL: Do analysts forecast differently in periods of uncertainty? An empirical analysis of target prices for Spanish banks.
- 2145 BEATRIZ GONZÁLEZ, GALO NUÑO, DOMINIK THALER and SILVIA ABRIZIO: Firm heterogeneity, capital misallocation and optimal monetary policy.
- 2201 RYAN BANERJEE and JOSÉ-MARÍA SERENA: Dampening the financial accelerator? Direct lenders and monetary policy.
- 2202 JUAN S. MORA-SANGUINETTI and ISABEL SOLER: La regulación sectorial en España. Resultados cuantitativos.
- 2203 JORGE E. GALÁN, MATÍAS LAMAS and RAQUEL VEGAS: Roots and recourse mortgages: handing back the keys.
- 2204 MONICA MARTINEZ-BRAVO and CARLOS SANZ: Inequality and psychological well-being in times of COVID-19: evidence from Spain.
- 2205 CORINNA GHIRELLI, DANILO LEIVA-LEÓN and ALBERTO URTASUN: Housing prices in Spain: convergence or decoupling?
- 2206 MARÍA BRU MUÑOZ: Financial exclusion and sovereign default: The role of official lenders.
- 2207 RICARDO GIMENO and CLARA I. GONZÁLEZ: The role of a green factor in stock prices. When Fama & French go green.
- 2208 CARLOS MONTES-GALDÓN and EVA ORTEGA: Skewed SVARs: tracking the structural sources of macroeconomic tail risks.
- 2209 RODOLFO G. CAMPOS, ILIANA REGGIO and JACOPO TIMINI: Thick borders in Franco's Spain: the costs of a closed economy.
- 2210 MARCO CELENTANI, MIGUEL GARCÍA-POSADA and FERNANDO GÓMEZ POMAR: Fresh start policies and small business activity: evidence from a natural experiment.
- 2211 JOSE GARCIA-LOUZAO, LAURA HOSPIDO and ALESSANDRO RUGGIERI: Dual returns to experience.
- 2212 ADRIÁN CARRO and PATRICIA STUPARIU: Uncertainty, non-linear contagion and the credit quality channel: an application to the Spanish interbank market.
- 2113 MARIO IZQUIERDO, ENRIQUE MORAL-BENITO, ELVIRA PRADES and JAVIER QUINTANA: The propagation of worldwide sector-specific shocks.
- 2214 HENRIQUE S. BASSO: Asset holdings, information aggregation in secondary markets and credit cycles.
- 2215 JOSÉ MANUEL CARBÓ and SERGIO GORJÓN: Application of machine learning models and interpretability techniques to identify the determinants of the price of bitcoin.
- 2216 LUIS GUIROLA and MARÍA SÁNCHEZ-DOMÍNGUEZ: Childcare constraints on immigrant integration.
- 2217 ADRIÁN CARRO, MARC HINTERSCHWEIGER, ARZU ULUC and J. DOYNE FARMER: Heterogeneous effects and spillovers of macroprudential policy in an agent-based model of the UK housing market.
- 2218 STÉPHANE DUPRAZ, HERVÉ LE BIHAN and JULIEN MATHERON: Make-up strategies with finite planning horizons but forward-looking asset prices.
- 2219 LAURA ÁLVAREZ, MIGUEL GARCÍA-POSADA and SERGIO MAYORDOMO: Distressed firms, zombie firms and zombie lending: a taxonomy.
- 2220 BLANCA JIMÉNEZ-GARCÍA and JULIO RODRÍGUEZ: A quantification of the evolution of bilateral trade flows once bilateral RTAs are implemented.
- 2221 SALOMÓN GARCÍA: Mortgage securitization and information frictions in general equilibrium.
- 2222 ANDRÉS ALONSO and JOSÉ MANUEL CARBÓ: Accuracy of explanations of machine learning models for credit decisions.
- 2223 JAMES COSTAIN, GALO NUÑO and CARLOS THOMAS: The term structure of interest rates in a heterogeneous monetary union.
- 2224 ANTOINE BERTHEAU, EDOARDO MARIA ACABBI, CRISTINA BARCELÓ, ANDREAS GULYAS, STEFANO LOMBARDI and RAFFAELE SAGGIO: The Unequal Consequences of Job Loss across Countries.
- 2225 ERWAN GAUTIER, CRISTINA CONFLITTI, RIEMER P. FABER, BRIAN FABO, LUDMILA FADEJEVA, VALENTIN JOUVANCEAU, JAN-OLIVER MENZ, TERESA MESSNER, PAVLOS PETROULAS, PAU ROLDAN-BLANCO, FABIO RUMLER, SERGIO SANTORO, ELISABETH WIELAND and HÉLÈNE ZIMMER. New facts on consumer price rigidity in the euro area.
- 2226 MARIO BAJO and EMILIO RODRÍGUEZ: Integrating the carbon footprint into the construction of corporate bond portfolios.
- 2227 FEDERICO CARRIL-CACCIA, JORDI PANIAGUA and MARTA SUÁREZ-VARELA: Forced migration and food crises.
- 2228 CARLOS MORENO PÉREZ and MARCO MINOZZO: Natural Language Processing and Financial Markets: Semi-supervised Modelling of Coronavirus and Economic News.
- 2229 CARLOS MORENO PÉREZ and MARCO MINOZZO: Monetary Policy Uncertainty in Mexico: An Unsupervised Approach.
- 2230 ADRIAN CARRO: Could Spain be less different? Exploring the effects of macroprudential policy on the house price cycle.
- 2231 DANIEL SANTABÁRBARA and MARTA SUÁREZ-VARELA: Carbon pricing and inflation volatility.